%\documentclass[Proceedings]{ascelike}
%\documentclass[Journal,BackFigs]{ascelike}
\documentclass[Journal,InsideFigs]{ascelike}
%\documentclass[Journal,InsideFigs]{ascelike}
\usepackage{graphicx}
%\usepackage{amsmath}
%\usepackage{natbib}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\NameTag{Sett et al.}
%
\begin{document}
\begin{thebibliography}{}

\bibitem[\protect\citeauthoryear{Anders and Hori}{Anders and
  Hori}{1999}]{Hori:1999}
Anders, M. and Hori, M. (1999).
\newblock ``Stochastic finite element method for elasto-plastic body.''\ {\em
  International Journal for Numerical Methods in Engineering}, 46, 1897--1916.

\bibitem[\protect\citeauthoryear{Anders and Hori}{Anders and
  Hori}{2000}]{Hori:2000}
Anders, M. and Hori, M. (2000).
\newblock ``Three-dimensional stochastic finite element method for
  elasto-plastic bodies.''\ {\em International Journal for Numerical Methods in
  Engineering}, 51, 449--478.

\bibitem[\protect\citeauthoryear{Baecher and Christian}{Baecher and
  Christian}{2003}]{book:Christian}
Baecher, G.~B. and Christian, J.~T. (2003).
\newblock {\em Reliability and Statistics in Geotechnical Engineering}.
\newblock Wiley, West Sussex PO19 8SQ, England, second edition.
\newblock ISBN 0-471-49833-5.

\bibitem[\protect\citeauthoryear{Ben-Haim and Elishakoff}{Ben-Haim and
  Elishakoff}{1990}]{book:Elishakoff}
Ben-Haim, Y. and Elishakoff, I. (1990).
\newblock {\em Convex Models of Uncertainty in Applied Mechanics}.
\newblock Elsevier, Amsterdam.

\bibitem[\protect\citeauthoryear{Borja}{Borja}{2004}]{borja:OpinionPaper}
Borja, R.~I. (2004).
\newblock ``Incorporating uncertainties in nonlinear soil properties into
  numerical models.''\ {\em Proceedings of International Workshop on
  Uncertainties in Nonlinear Soil Properties and their Impact on Modeling
  Dynamic Soil Response, PEER Headquarters, UC Berkeley, March 18-19, 2004},
  Opinion Paper, PEER,
  http://peer.berkeley.edu/lifelines/Workshop304/pdf/o{\_}Borja.pdf.

\bibitem[\protect\citeauthoryear{De~Lima, Teixeira, and Ebecken}{De~Lima
  et~al.}{2001}]{DeLima:SFEMgeotechApplication}
De~Lima, B. S. L.~P., Teixeira, E.~C., and Ebecken, N. F.~F. (2001).
\newblock ``Probabilistic and possibilistic methods for the elastoplastic
  analysis of soils.''\ {\em Advances in Engineering Software}, 132, 569--585.

\bibitem[\protect\citeauthoryear{DeGroot and Baecher}{DeGroot and
  Baecher}{1993}]{Baecher:1993}
DeGroot, D.~J. and Baecher, G.~B. (1993).
\newblock ``Estimating autocovariance of in-situ soil properties.''\ {\em
  Journal of Geotechnical Engineering}, 119(1), 147--166.

\bibitem[\protect\citeauthoryear{Duncan}{Duncan}{2000}]{Duncan:2000}
Duncan, J.~M. (2000).
\newblock ``Factors of safety and reliability in geotechnical engineering.''\
  {\em Journal of Geotechnical and Geoenvironmental Engineering, ASCE}, 126(4),
  307--316.

\bibitem[\protect\citeauthoryear{Einav and Collins}{Einav and
  Collins}{2008}]{Einav:2008}
Einav, I. and Collins, I.~F. (2008).
\newblock ``A thermomechanical framework of plasticity based on probabilistic
  micromechanics.''\ {\em Journal of Mechanics of Materials and Structures},
  3(5), 867--892.

\bibitem[\protect\citeauthoryear{{Federal Highway Administration, U.S.
  Department of Transportation}}{{Federal Highway Administration, U.S.
  Department of Transportation}}{2002}]{FHWA:2002}
{Federal Highway Administration, U.S. Department of Transportation} (2002).
\newblock ``Geotechnical engineering circular no. 5.
\newblock Publication No. FHWA-IF-02-034.

\bibitem[\protect\citeauthoryear{Fenton}{Fenton}{1999a}]{Fenton:1999A}
Fenton, G.~A. (1999a).
\newblock ``Estimation of stochastic soil models.''\ {\em Journal of
  Geotechnical and Geoenvironmental Engineering, ASCE}, 125(6), 470--485.

\bibitem[\protect\citeauthoryear{Fenton}{Fenton}{1999b}]{Fenton:1999B}
Fenton, G.~A. (1999b).
\newblock ``Random field modeling of {CPT} data.''\ {\em Journal of
  Geotechnical and Geoenvironmental Engineering, ASCE}, 125(6), 486--498.

\bibitem[\protect\citeauthoryear{Fenton and Griffiths}{Fenton and
  Griffiths}{2002}]{Griffiths:2002a}
Fenton, G.~A. and Griffiths, D.~V. (2002).
\newblock ``Probabilistic foundation settlement on spatially random soil.''\
  {\em Journal of Geotechnical and Geoenvironmental Engineering, ASCE}, 128(5),
  381--390.

\bibitem[\protect\citeauthoryear{Fenton and Griffiths}{Fenton and
  Griffiths}{2003}]{Griffiths:2003}
Fenton, G.~A. and Griffiths, D.~V. (2003).
\newblock ``Bearing capacity prediction of spatially random $c-\phi$ soil.''\
  {\em Canadian Geotechnical Journal}, 40, 54--65.

\bibitem[\protect\citeauthoryear{Fenton and Griffiths}{Fenton and
  Griffiths}{2005}]{Griffiths:2005}
Fenton, G.~A. and Griffiths, D.~V. (2005).
\newblock ``Three-dimensional probabilistic foundation settlement.''\ {\em
  Journal of Geotechnical and Geoenvironmental Engineering, ASCE}, 131(2),
  232--239.

\bibitem[\protect\citeauthoryear{Gardiner}{Gardiner}{2004}]{book:Gardiner}
Gardiner, C.~W. (2004).
\newblock {\em Handbook of Stochastic Methods for Physics, Chemistry and the
  Natural Science}.
\newblock Springer:Complexity. Springer-Verlag, Berlin Heidelberg, third
  edition.

\bibitem[\protect\citeauthoryear{Ghanem and Spanos}{Ghanem and
  Spanos}{1991}]{book:Ghanem}
Ghanem, R.~G. and Spanos, P.~D. (1991).
\newblock {\em Stochastic Finite Elements: A Spectral Approach}.
\newblock Springer-Verlag.
\newblock (Reissued by Dover Publications, 2003).

\bibitem[\protect\citeauthoryear{Griffiths, Fenton, and Manoharan}{Griffiths
  et~al.}{2002}]{Griffiths:2002}
Griffiths, D.~V., Fenton, G.~A., and Manoharan, N. (2002).
\newblock ``Bearing capacity of rough rigid strip footing on cohesive soil:
  Probabilistic study.''\ {\em Journal of Geotechnical and Geoenvironmental
  Engineering, ASCE}, 128(9), 743--755.

\bibitem[\protect\citeauthoryear{Hammitt}{Hammitt}{1966}]{Hammitt:1966}
Hammitt, G.~M. (1966).
\newblock ``Statistical analysis of data from a comparative laboratory test
  program sponsored by {ACITL}.
\newblock U.S. Army Waterways Experiment Station, Vicksburg, MS.

\bibitem[\protect\citeauthoryear{Hara, Ohta, Niwa, Tanaka, and Banno}{Hara
  et~al.}{1974}]{Hara:1974}
Hara, A., Ohta, T., Niwa, M., Tanaka, S., and Banno, T. (1974).
\newblock ``Shear modulus and shear strength of cohesive soils.''\ {\em Soils
  and Foundations}, 14(3), 1--12.

\bibitem[\protect\citeauthoryear{Harr}{Harr}{1987}]{book:Harr}
Harr, M.~E. (1987).
\newblock {\em Reliability-Based Design in Civil Engineering}.
\newblock McGraw--Hill.

\bibitem[\protect\citeauthoryear{Hasancebi and Ulusay}{Hasancebi and
  Ulusay}{2007}]{Hasancebi:2007}
Hasancebi, N. and Ulusay, R. (2007).
\newblock ``Emperical correlations between shear wave velocity and penetration
  resistance for ground shaking assessments.''\ {\em Bulletin of Engineering
  Geology and Environment}, 66(2), 203--213.

\bibitem[\protect\citeauthoryear{Hashin}{Hashin}{1983}]{local-34}
Hashin, Z. (1983).
\newblock ``Analysis of composite materials.''\ {\em Journal of Applied
  Mechanics}, 50, 481--505.

\bibitem[\protect\citeauthoryear{Idriss}{Idriss}{1990}]{Idriss:1990}
Idriss, I.~M. (1990).
\newblock ``Response of soft soil sites during earthquakes.''\ {\em Proceedings
  of the Symposium to Honor Professor H. B. Seed}.  273--289.

\bibitem[\protect\citeauthoryear{Imai}{Imai}{1977}]{Imai:1977}
Imai, T. (1977).
\newblock ``{P}- and {S}-wave velocities of the ground in {J}apan.''\ {\em
  Proceedings of the IX International Conference on Soil Mechanics and
  Foundation Engineering}.  127--132.
\newblock Volume 2.

\bibitem[\protect\citeauthoryear{Jafari, Shafiee, and Ramzkhah}{Jafari
  et~al.}{2002}]{Jafari:2002}
Jafari, M.~K., Shafiee, A., and Ramzkhah, A. (2002).
\newblock ``Dynamic properties of the fine grained soils in south of
  {T}ehran.''\ {\em Iranian Journal of Seismology and Earthquake Engineering},
  4(1), 25--35.

\bibitem[\protect\citeauthoryear{Jeremi{\'c} and Sett}{Jeremi{\'c} and
  Sett}{2009}]{Jeremic:2009b}
Jeremi{\'c}, B. and Sett, K. (2009).
\newblock ``On probabilistic yielding of materials.''\ {\em Communications in
  Numerical Methods in Engineering}, 25(3), 291--300.

\bibitem[\protect\citeauthoryear{Jeremi{\'c}, Sett, and Kavvas}{Jeremi{\'c}
  et~al.}{2007}]{Jeremic:2007}
Jeremi{\'c}, B., Sett, K., and Kavvas, M.~L. (2007).
\newblock ``Probabilistic elasto-plasticity: Formulation in {1--D}.''\ {\em
  Acta Geotechnica}, 2(3), 197--210.

\bibitem[\protect\citeauthoryear{Karhunen}{Karhunen}{1947}]{Karhunen:1947}
Karhunen, K. (1947).
\newblock ``{\"U}ber lineare methoden in der wahrscheinlichkeitsrechnung.''\
  {\em Ann. Acad. Sci. Fennicae. Ser. A. I. Math.-Phys.}, (37), 1--79.

\bibitem[\protect\citeauthoryear{Kavvas}{Kavvas}{2003}]{Kavvas:2003}
Kavvas, M.~L. (2003).
\newblock ``Nonlinear hydrologic processes: Conservation equations for
  determining their means and probability distributions.''\ {\em Journal of
  Hydrologic Engineering}, 8(2), 44--53.

\bibitem[\protect\citeauthoryear{Kleiber and Hien}{Kleiber and
  Hien}{1992}]{book:Kleiber}
Kleiber, M. and Hien, T.~D. (1992).
\newblock {\em The Stochastic Finite Element Method: Basic Perturbation
  Technique and Computer Implementation}.
\newblock John Wiley {\&} Sons, Baffins Lane, Chichester, West Sussex PO19 1UD
  , England.

\bibitem[\protect\citeauthoryear{Kulhawy and Phoon}{Kulhawy and
  Phoon}{2002}]{Kulhawy:2002}
Kulhawy, F.~H. and Phoon, K.~K. (2002).
\newblock ``Observations on geotechnical reliability-based design development
  in {N}orth {A}merica.''\ {\em Foundation Design Codes and Soil Investigation
  in view of International Harmonization and Performance Based Design}, Y.
  Honjo, O. Kusakabe, M. Kenji, M. Kouda, and G. Pokharel, eds. Swets and
  Zeitlinger, Lisse,  31--48.

\bibitem[\protect\citeauthoryear{Lacasse and Nadim}{Lacasse and
  Nadim}{1996}]{Lacasse:1996}
Lacasse, S. and Nadim, F. (1996).
\newblock ``Uncertainties in characterizing soil properties.''\ {\em
  Uncertainty in Geologic Environment: From Theory to Practice, Proceedings of
  Uncertainty '96, July 31-August 3, 1996, Madison, Wisconsin}, C.~D.
  Shackelford and P.~P. Nelson, eds., Vol.~1 of {\em Geotechnical Special
  Publication No. 58}. ASCE, New York,  49--75.

\bibitem[\protect\citeauthoryear{Ladd}{Ladd}{1991}]{Ladd:1991}
Ladd, C.~C. (1991).
\newblock ``Stability evaluation during staged construction.''\ {\em Journal of
  Geotechnical Engineering}, 117(4), 540--615.

\bibitem[\protect\citeauthoryear{Lo{\`e}ve}{Lo{\`e}ve}{1948}]{Loeve:1948}
Lo{\`e}ve, M. (1948).
\newblock ``Fonctions aleatories du second ordre.
\newblock Supplement to P. Levy, Processus Stochastic et Mouvement Brownien,
  Gauthier-Villars, Paris.

\bibitem[\protect\citeauthoryear{Lumb}{Lumb}{1966}]{Lumb:1966}
Lumb, P. (1966).
\newblock ``The variability of natural soils.''\ {\em Canadian Geotechnical
  Journal}, 3, 74--97.

\bibitem[\protect\citeauthoryear{Manolis}{Manolis}{2002}]{Manolis:2002}
Manolis, G.~D. (2002).
\newblock ``Stochastic soil dynamics.''\ {\em Soil Dynamics and Earthquake
  Engineering}, 22, 3--15.
\newblock Review Paper.

\bibitem[\protect\citeauthoryear{Marosi and Hiltunen}{Marosi and
  Hiltunen}{2004}]{Marosi:2004}
Marosi, K.~T. and Hiltunen, D.~R. (2004).
\newblock ``Characterization of spectral analysis of surface waves shear wave
  velocity measurement uncertainty.''\ {\em Journal of Geotechnical and
  Geoenvironmental Engineering, ASCE}, 130(10), 1034--1041.

\bibitem[\protect\citeauthoryear{Matthies, Brenner, Bucher, and
  Soares}{Matthies et~al.}{1997}]{Matthies:1997reviewSFEM}
Matthies, H.~G., Brenner, C.~E., Bucher, C.~G., and Soares, C.~G. (1997).
\newblock ``Uncertainties in probabilistic numerical analysis of structures and
  soilds - stochastic finite elements.''\ {\em Structural Safety}, 19(3),
  283--336.

\bibitem[\protect\citeauthoryear{Mayne}{Mayne}{2007}]{Mayne:2007}
Mayne, P.~W. (2007).
\newblock ``Cone penetration testing state-of-practice.''\ {\em Final Report
  NCHRP Project 20-05; Task 37-14}, Georgia Institute of Technology, Atlanta,
  GA.
\newblock Transportation Research Board Synthesis Study.

\bibitem[\protect\citeauthoryear{Metropolis and Ulam}{Metropolis and
  Ulam}{1949}]{Ulam:1949}
Metropolis, N. and Ulam, S. (1949).
\newblock ``The {M}onte {C}arlo method.''\ {\em Journal of the American
  Statistical Association}, 44, 335--341.

\bibitem[\protect\citeauthoryear{Moore}{Moore}{1979}]{book:Moore}
Moore, R. (1979).
\newblock {\em Methods and Applications of Interval Analysis}.
\newblock SIAM, Philadelphia.

\bibitem[\protect\citeauthoryear{Ohya, Imai, and Matsubara}{Ohya
  et~al.}{1982}]{Ohya:1982}
Ohya, S., Imai, T., and Matsubara, M. (1982).
\newblock ``Relationship between {N}-value by {SPT} and {LLT} pressuremeter
  results.''\ {\em Proceedings of the 2nd. European Symposium on Penetration
  Testing}, Vol.~1, Amsterdam.  125--130.

\bibitem[\protect\citeauthoryear{Paice, Griffiths, and Fenton}{Paice
  et~al.}{1996}]{Griffiths:1996}
Paice, G.~M., Griffiths, D.~V., and Fenton, G.~A. (1996).
\newblock ``Finite element modeling of settlement on spatially random soil.''\
  {\em Journal of Geotechnical Engineering}, 122(9), 777--779.

\bibitem[\protect\citeauthoryear{Phoon and Kulhawy}{Phoon and
  Kulhawy}{1999a}]{Kulhawy:1999A}
Phoon, K.-K. and Kulhawy, F.~H. (1999a).
\newblock ``Characterization of geotechnical variability.''\ {\em Canadian
  Geotechnical Journal}, 36, 612--624.

\bibitem[\protect\citeauthoryear{Phoon and Kulhawy}{Phoon and
  Kulhawy}{1999b}]{Kulhawy:1999B}
Phoon, K.-K. and Kulhawy, F.~H. (1999b).
\newblock ``Evaluation of geotechnical property variability.''\ {\em Canadian
  Geotechnical Journal}, 36, 625--639.

\bibitem[\protect\citeauthoryear{Pitilakis, Raptakis, Lontzetidis,
  Tika-Vasilikou, and Jongmans}{Pitilakis et~al.}{1999}]{Pitilakis:1999}
Pitilakis, K., Raptakis, D., Lontzetidis, K., Tika-Vasilikou, T., and Jongmans,
  D. (1999).
\newblock ``Geotechnical and geophysical description of {E}uro-seistests, using
  field and laboratory tets, and moderate strong ground motions.''\ {\em
  Journal of Earthquake Engineering}, 3(3), 381--409.

\bibitem[\protect\citeauthoryear{Popescu, Deodatis, and Nobahar}{Popescu
  et~al.}{2005}]{Popescu:2005}
Popescu, R., Deodatis, G., and Nobahar, A. (2005).
\newblock ``Effects of random heterogeneity of soil properties on bearing
  capacity.''\ {\em Probabilistic Engineering Mechanics}, 20, 324--341.

\bibitem[\protect\citeauthoryear{Rackwitz}{Rackwitz}{2000}]{Rackwitz:2000}
Rackwitz, R. (2000).
\newblock ``Reviewing probabilistic soil modelling.''\ {\em Computers and
  Geotechnics}, 26, 199--223.

\bibitem[\protect\citeauthoryear{Sett and Jeremi{\'c}}{Sett and
  Jeremi{\'c}}{2009}]{Sett:2009b}
Sett, K. and Jeremi{\'c}, B. (2009).
\newblock ``Forward and backward probabilistic simulations in geotechnical
  engineering.''\ {\em Contemporary Topics in Insitu Testing, Analysis, and
  Reliability of Foundations (Selected Papers from the 2009 International
  Foundation Congress and Equipment Expo, Orlando, FL, March 15-19, 2009)}, M.
  Iskander, D.~F. Laefer, and M.~H. Hussein, eds., Geotechnical Special
  Publications No. 186. American Society for Civil Engineers,  1--11.

\bibitem[\protect\citeauthoryear{Sett and Jeremi{\'c}}{Sett and
  Jeremi{\'c}}{2010}]{Sett:2010a}
Sett, K. and Jeremi{\'c}, B. (2010).
\newblock ``Probabilistic yielding and cyclic behavior of geomaterials.''\ {\em
  International Journal for Numerical and Analytical Methods in Geomechanics}.
\newblock in print (published online in the Early View Section, DOI:
  10.1002/nag.870).

\bibitem[\protect\citeauthoryear{Sett, Jeremi{\'c}, and Kavvas}{Sett
  et~al.}{2007a}]{Sett:2007b}
Sett, K., Jeremi{\'c}, B., and Kavvas, M.~L. (2007a).
\newblock ``Probabilistic elasto-plasticity: Solution and verification in
  {1--D}.''\ {\em Acta Geotechnica}, 2(3), 211--220.

\bibitem[\protect\citeauthoryear{Sett, Jeremi{\'c}, and Kavvas}{Sett
  et~al.}{2007b}]{sett:2007a}
Sett, K., Jeremi{\'c}, B., and Kavvas, M.~L. (2007b).
\newblock ``The role of nonlinear hardening/softening in probabilistic
  elasto--plasticity.''\ {\em International Journal for Numerical and
  Analytical Methods in Geomechanics}, 31(7), 953--975.

\bibitem[\protect\citeauthoryear{Sett, Jeremi{\'c}, and Kavvas}{Sett
  et~al.}{2010}]{Sett:2010c}
Sett, K., Jeremi{\'c}, B., and Kavvas, M.~L. (2010).
\newblock ``Stochastic elastic-plastic finite elements.''\ {\em Computer
  Methods in Applied Mechanics and Engineering}.
\newblock In review.

\bibitem[\protect\citeauthoryear{Stokoe~II, Darendeli, Gilbert, Menq, and
  Choi}{Stokoe~II et~al.}{2004}]{Stokoe:2004}
Stokoe~II, K.~H., Darendeli, R.~B., Gilbert, R.~B., Menq, F.-Y., and Choi,
  W.~K. (2004).
\newblock ``Development of a new family of normalized modulus reduction and
  material damping curves.''\ {\em International Workshop on Uncertainties in
  Nonlinear Soil Properties and their Impact on Modeling Dynamic Soil Response,
  PEER Headquarters, UC Berkeley, March 18-19, 2004}, Plenary Paper, PEER,
  http://peer.berkeley.edu/lifelines/
  Workshop304/pdf/Stokoe{\_}PlenaryPaper.pdf.

\bibitem[\protect\citeauthoryear{Sudret and Der~Kiureghian}{Sudret and
  Der~Kiureghian}{2000}]{DerKiureghian:2000ReportSFEM}
Sudret, B. and Der~Kiureghian, A. (2000).
\newblock ``Stochastic finite element methods and reliability: A state of the
  art report.''\ {\em Technical Report UCB/SEMM-2000/08}, University of
  California, Berkeley.

\bibitem[\protect\citeauthoryear{Vucetic and Dobry}{Vucetic and
  Dobry}{1991}]{Vucetic:1991}
Vucetic, M. and Dobry, R. (1991).
\newblock ``Effect of soil plasticity on cyclic response.''\ {\em Journal of
  Geotechnical Engineering}, 117(1), 133--149.

\bibitem[\protect\citeauthoryear{Wiener}{Wiener}{1938}]{Wiener:1938}
Wiener, N. (1938).
\newblock ``The homogeneous chaos.''\ {\em American Journal of Mathematics},
  60, 897--936.

\bibitem[\protect\citeauthoryear{Zadeh}{Zadeh}{1983}]{Zadeh:1983}
Zadeh, J. (1983).
\newblock ``The role of fuzzy logic in the management of uncertainty in expert
  systems.''\ {\em Fuzzy Sets and Systems}, 11, 199--228.

\end{thebibliography}


\title{Soil Uncertainty and its Influence on Simulated $G/G_{max}$ and Damping Behavior}

\author{
Kallol Sett\thanks{
Department        of    Civil  Engineering, The University of Akron, Akron, Ohio, U.S.A.},\ A.M. ASCE,
%
Berna Unutmaz\thanks{
Department of Civil Engineering, Kocaeli University, Kocaeli, Turkey},
%
Kemal \"{O}nder \c{C}etin\thanks{
Department of Civil Engineering, Middle East Technical University, Ankara, Turkey},\ M. ASCE, \\ 
%
Suzana Koprivica\thanks{
Department of Civil Construction Management, Union University, Belgrade, Serbia},
%
and 
Boris Jeremi{\'c}\thanks{
Corresponding author; Department of Civil  and  Environmental  Engineering, University of California, Davis, California, U.S.A.},\ M. ASCE
}

\maketitle

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\ \\

\noindent
\small{\bf To be published in the \\
 ASCE Journal of Geotechnical and Geoenvironmental Engineering, 2010.}

\noindent \textbf{\uppercase{Abstract:}} In   this   paper,   recently   developed  probabilistic
elasto--plasticity is applied in simulating cyclic 
behavior of clay. Simple von Mises elastic--perfectly plastic material
model  is  used  for simulation. Probabilistic soil parameters, elastic
shear   modulus   ($G_{max}$)   and  undrained  shear  strength
($s_u$), that are needed   for   the  simulation  are  obtained  from
correlations  with  SPT  $N$-value.  It  has been shown that the
probabilistic approach to geomaterial modeling captures some of
the  important  aspects  -- modulus reduction, material damping
ratio,  and  modulus degradation -- of cyclic behavior of clay
reasonably well, even with the simple elastic--perfectly
plastic material model.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{\uppercase{Introduction}}

Behavior  of geomaterials  is  inherently uncertain. This uncertainty stems from
natural soil variability, testing and transformation errors
\cite{Lacasse:1996,Kulhawy:1999A}.
%
Traditionally,  geotechnical  engineering  community deals with
uncertainties  in  geomaterial  by  applying   factor  of
safety. However, use of (large) factors of safety not only result
in  over-expensive  design,  but also, sometimes, in unsafe design
\cite{Duncan:2000}.    Hence,   in   recent   years,
geotechnical   engineering  practice  has  seen  an  increasing
emphasis  on  probabilistic  treatment  to  data and subsequent
simulation/design.


Quantification  or  mathematical  description of uncertainty is
usually  done  within  the  framework  of  probability  theory,
although    fuzzy   sets   \cite{Zadeh:1983},   convex   models
\cite{book:Elishakoff},  and interval  arithmetic \cite{book:Moore}
have  also  been  used  in  the  past  to  describe uncertainty
mathematically.  
%
Under   the  framework  of  probability theory, uncertain  spatial  variability 
of  soil  deposit is modeled as random field -- a collection of
random variables, indexed from space continuum.
%
For complete characterization (up to second order) of a random field, 
in addition to mean and variance, information on
autocovariance function and correlation
length  or  scale of fluctuation are also needed.  
%
Thorough  descriptions  of  random  field  modeling
techniques,  with  procedures  for estimating correlation length
for    geotechnical  engineering applications, are given by  \citeN{Baecher:1993}
and \citeN{Fenton:1999A}.     
%
Testing     and     transformation
uncertainties, on the other hand,  are  
point uncertainties and are usually modeled as random
variables, which are completely characterized (up to
second order) by their respective means and variances.
%
%
Over     the    years    researchers    have  quantified  and  collected typical
variations  of different soil properties, ranging from consolidation parameters,
laboratory     measured    strength    properties    to    in-situ    properties
\cite{Lumb:1966,Lacasse:1996,book:Christian,Kulhawy:1999A}, as well as testing
uncertainties, associated with the most commonly used test methods \cite{Hammitt:1966,Kulhawy:1999A,Marosi:2004},
and transformation uncertainties, associated with the most common
transformation relationships \cite{Kulhawy:1999B}.
%
A fair amount of work was also done
on  subsequent  probabilistic  geotechnical  design  guidelines
\cite{book:Harr,Kulhawy:2002}, although   
the existing geotechnical
LRFD codes still do not explicitly consider the soil properties
uncertainties.
%
The   book  by
\citeN{book:Christian}   thoroughly
describes   the   current   state-of-the-art  of  probabilistic
geotechnical  engineering design. 


Modeling  and  simulation under uncertainty, 
on the other hand,  have received much less attention, mainly  due to the concern about the the necessity,
usefulness,  and  tractability  of  probabilistic  modeling  in
geotechnical   engineering,   when  geotechnical  problems  are
difficult  to  model  even  deterministically,  unless advanced
modeling techniques are used.
%
However, published works on this subject \cite{Griffiths:1996,Rackwitz:2000,Manolis:2002,Griffiths:2002,Griffiths:2002a,DeLima:SFEMgeotechApplication,Griffiths:2003,borja:OpinionPaper,Griffiths:2005,Popescu:2005} show very promising
results, especially in quantifying our confidence in our simulation.
%
Most of the above works are based on Monte Carlo technique \cite{Ulam:1949} in tandem with (deterministic) finite element
method. 
%
Monte Carlo technique relies on repeated random sampling (of, for example, soil properties) and because of this
repeatability, the computational cost associated with it could become extremely huge (and probably intractable), especially for
large scale problems, like dynamic soil-structure interaction analysis.
%
Due to the above drawback of Monte Carlo technique, in other fields of science and engineering,
stochastic differential equation approach \cite{book:Gardiner} or numerically, stochastic finite element method 
\cite{book:Kleiber,book:Ghanem} is very popular. However, non-linearities in soil properties prevent direct application of 
those techniques for probabilistic simulations in geotechnical engineering.


The difficulty in propagating uncertainties in soil properties through
the elastic-plastic constitutive equation lies in the high non-linear
dependence of the elastic-plastic modulus on stress.
%
Very  few published literature exist on this subject.
%
In fact, the first attempt to propagate uncertainties
through elastic-plastic constitutive rate equation was
published only recently \cite{Hori:1999}.
%
Anders and Hori \citeyear{Hori:1999,Hori:2000} used perturbation
approach -- a linearized Taylor series expansion
with respect to mean -- in propagating uncertainties
in modulus through the von Mises elastic--perfectly
material model.
%
However, inherent to the Taylor series expansion, perturbation
technique suffers from small variance requirement \cite{Matthies:1997reviewSFEM}.
%
A rule of thumb restricts the coefficient of variance (COV) to 20\% \cite{DerKiureghian:2000ReportSFEM} to minimize the error
in perturbation approach. 
%
This severely limits the applicability of perturbation approach to geotechnical
problems, where soil COVs are rarely less than 20\% (cf. Phoon and Kulhawy \citeyear{Kulhawy:1999A,Kulhawy:1999B}).
%
Perturbation approach also suffers from closure problem \cite{Kavvas:2003}, which means that
information about higher-order statistical moments are necessary to
calculate lower-order statistical moments. 
%
Griffiths and Fenton \citeyear{Griffiths:2002a,Griffiths:2003,Griffiths:2005}
used brute force Monte Carlo technique in propagating uncertainties through 
elastic-perfectly plastic Mohr-Coulomb model.
%
Recently, in circumventing the above drawbacks of Monte Carlo and
perturbation techniques, \citeN{Jeremic:2007} proposed Eulerian--Lagrangian
Fokker--Planck--Kolmogorov  equation (FPKE; \citeNP{Kavvas:2003}) based
probabilistic   elasto--plasticity.
%
Developed probabilistic   elasto--plasticity is  compatible with the general
theory of (deterministic) plasticity, and hence can be used 
for   probabilistic   simulation of a variety of elastic--plastic   models. 
%
Solution strategies for probabilistic elasto--plasticity were discussed by 
Sett et al. \citeyear{Sett:2007b,sett:2007a} for both linear and non-linear
hardening  models.  The  concept  of probabilistic yielding was
introduced by \citeN{Jeremic:2009b}, while  \citeN{Sett:2010a}
discussed  its  effect  on constitutive simulation under cyclic
loading.  
%
It was shown that due to uncertainty in yield stress,
there  is always a possibility that  plastic  behavior  starts  at very very low
strain  and  influence  of  elastic behavior continues far into the
plastic  domain. 
%
Because of this, the average  (mean)  constitutive response and the most probable
(mode)  constitutive response  show  nonlinear behavior with a vanishing
linear region, even for the simplest elastic--perfectly plastic
material  model.  
%
This  in turn is significant since, by expanding into probability   space,  one 
obtains  very realistic soil behavior with simple constitutive models, requiring
very  few  soil  parameters  (and  their  distributions)  that  could be obtained
directly from in--situ tests (e.g. SPT, CPT etc.).
%

In  this  paper, FPKE based probabilistic approach to geomaterial modeling
is  applied  in simulating $G/G_{max}$ and damping behavior of undrained
clay.  Elastic--perfectly  plastic  von  Mises  material  model, which requires only two soil
parameters, the shear  modulus  ($G_{max}$)  and  the undrained shear
strength   ($s_u$), is used. 
%
Simulated responses
are  then  compared with published experimental measurements. 


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\clearpage
\section{\uppercase{Probabilistic Framework for Constitutive Simulation}}
\label{PROBABILISTIC_FRAMEWORK}

The   constitutive   behavior   of  soil  can  be  modeled  by  elastic--plastic
constitutive rate equation, which, in 1--D,  can be written
as:
%
\begin{equation}
\label{eqno_1}
\displaystyle \frac{d \sigma}{dt} = D \displaystyle \frac{d \epsilon}{dt}
\end{equation}
%
where    $\sigma$ is the stress, $\epsilon$  is  the  strain,  $t$  is  the  pseudo time, and $D$ is the
stiffness modulus, that can be either elastic or elastic--plastic:
%
\begin{eqnarray}
D = \left\{\begin{array}{ll}
%
           D^{el}
%
%
        \;\;\; & \mbox{elastic} \\
%
           \\
%
           D^{el}
           -
           \displaystyle \displaystyle \frac{D^{el}
           \displaystyle \frac{\partial U}{\partial \sigma}
           \displaystyle \frac{\partial f}{\partial \sigma}
           D^{el}}
           {\displaystyle \frac{\partial f}{\partial \sigma}
           D^{el}
           \displaystyle \frac{\partial U}{\partial \sigma}
           -
           \displaystyle \frac{\partial f}{\partial q_*}r_*}
%
        \;\;\; & \mbox{elastic-plastic}
%
           \end{array} \right.
\label{eqno_2}
\end{eqnarray}
%
where,  $D^{el}$,  $f$,  $U$,  $q_*$,  and $r_*$ are
elastic  modulus,  yield function,  plastic  potential function,
internal  variable(s),  and  rate(s)  of  evolution of internal
variable(s) respectively.  
%
However, due to various uncertainties
associated with soil properties, as discussed in the previous section, 
the modulus, $D$ in Eq.~(\ref{eqno_1}) becomes uncertain. 
%
In  traditional  deterministic  approach to elastic--plastic geo-material modeling, one typically
applies  engineering  judgment (qualitative) in obtaining the 'most probable' soil
parameters  and  substitute them  in Eq.~(\ref{eqno_1}) in obtaining the 'most probable'
soil constitutive behavior.
%
However, one may note that due to the non-linearity of soil behavior, 
'most probable' soil parameters do not necessarily result in 'most probable' 
constitutive behavior.



Recently,   \citeN{Jeremic:2007}   developed   a   probabilistic   approach  for
elastic--plastic  modeling  of  geomaterials.  Proposed approach is based on the
extension of constitutive
rate equation (Eq.~(\ref{eqno_1})) into probability density space using
Eulerian--Lagrangian Fokker--Planck--Kolmogorov approach \cite{Kavvas:2003}:
% 
% 
\begin{equation}
\label{eqno_3}
\displaystyle \frac{\partial P(\sigma(t), t)}{\partial t} =
-\displaystyle \frac{\partial}{\partial \sigma} \left\{ N_{(1)} P(\sigma(t),t) \right\}
+ \displaystyle \frac{\partial^2}{\partial \sigma^2} \left\{ N_{(2)} P(\sigma(t),t) \right\}
\end{equation}
%
% 
where $P(\sigma(t), t)$ is the probability density of
stress,  $t$  is  the  pseudo-time, while $N_{(1)}$ and $N_{(2)}$ are
advection   and   diffusion   coefficients,   respectively.  The
advection    and    diffusion  coefficients  depend only  on the material  model
used for modeling. 
%
It  can  be shown (see appendix of \citeNP{Jeremic:2007}), that the advection and
diffusion  coefficients  for  von  Mises  elastic-perfectly plastic constitutive
relationship    (used here to model undrained behavior of clay) can be
written as:
%  
%
\begin{eqnarray}
\begin{array}{l}
{}^{vM}\!N_{(1)}^{{el}} = \displaystyle \frac{d \epsilon}{dt} \left< G \right >
%
\;\;\;\;\;\; \mbox{;} \;\;\;\;\;\;
%
%\label{eqno_E2}
{}^{vM}\!N_{(2)}^{{el}}  =   t \left(\displaystyle \frac{d \epsilon}{dt}\right)^2 Var[G] 
\\
{}^{vM}\!N_{(1)}^{{el-pl}} = 0
\;\;\;\;\;\;\;\;\;\;\;\; \mbox{;} \;\;\;\;\;\;\;
{}^{vM}\!N_{(2)}^{{el-pl}} = 0 \;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;\;
\end{array}
\label{eqno_4}
\end{eqnarray}
%
%%
In    Eq.~(\ref{eqno_4}), $G$  is  the  elastic shear modulus and $\epsilon$ is
the shear strain. Furthermore, $\left< \cdot \right>$ represents the expectation
operator, while $Var\left[ \cdot \right]$ is the variance operator.
%
The  superscripts    $\cdot^{el}$    and    $\cdot^{el-pl}$  on  the  advection and
diffusion  coefficients  refer  to  pre--yield  elastic  region  and post-yield,
elastic--plastic region.
%
%

However,  for  a  heterogeneous  material  like  soil,  yield  strength is quite
uncertain. 
%
This is due to the fact that in a representative volume element (RVE; 
\citeNP{local-34}) of the heterogeneous
material,  each  of  the  large  number of particle contacts has different yield
strengths and orientations.
%
Each of these particle contacts will yield differently, depending upon its yield strength.
%
Hence, for  material  with  uncertain yield strength ($\Sigma_y$),
there exist possibilities that the material (RVE) starts yielding inside the elastic regime and/or
elastic behavior continues way into the plastic regime.
%
Under the framework of probability theory, these possibilities are
governed by the probability density function of yield strength ($\Sigma_y$), which
can be quantified by statistically analyzing the test results. 
%
Hence, to realistically simulate the probabilistic material behavior, \citeN{Jeremic:2009b}  
suggested  probability  weights, based on probability density function of yield 
strength  ($\Sigma_y$),  to  the  elastic  and  plastic  advection and diffusion
coefficients in obtaining equivalent advection and diffusion coefficients.
%
For  von  Mises  elastic--perfectly  plastic  soil with uncertain yield strength
($\Sigma_y$),    the    equivalent    advection   and   diffusion   coefficients
(${}^{vM}\!N_{(1)}^{{eq}}$    and    ${}^{vM}\!N_{(2)}^{{eq}}$)    would   become   (cf.
\citeNP{Sett:2010a}):
%
\begin{eqnarray}
%\begin{array}{l}
%\label{eqno_6a_VM}
\nonumber
{}^{vM}\!N_{(1)}^{{eq}} (\sigma) 
&=& (1 - P[\Sigma_y \leq \sigma]) {}^{vM}\!N_{(1)}^{{el}} + P[\Sigma_y \leq \sigma] {}^{vM}\!N_{(1)}^{{el-pl}} \\
\label{eqno_6a_VM}
&=& (1 - P[\Sigma_y \leq \sigma]) 
\displaystyle \frac{d \epsilon}{dt}
\left< G \right >
\ \\
\nonumber
\ \\
\nonumber
{}^{vM}\!N_{(2)}^{{eq}} (\sigma) 
&=& (1 - P[\Sigma_y \leq \sigma]) {}^{vM}\!N_{(2)}^{{el}}  + P[\Sigma_y \leq \sigma] {}^{vM}\!N_{(2)}^{{el-pl}} \\ 
\label{eqno_6b_VM}
&=& (1 - P[\Sigma_y \leq \sigma])
t \left( \displaystyle \frac{d \epsilon}{dt}\right)^2 Var[G] 
%\end{array}
\end{eqnarray}
%
The  probability  weight  ($  P[\Sigma_y  \leq  \sigma]$) in the above equations
(Eqs.~(\ref{eqno_6a_VM}) and (\ref{eqno_6b_VM})), quantifies, at any given stress level,
the  probability  of the material RVE being elastic or elastic--plastic. Using
the         above        equivalent        advection        and        diffusion
coefficients (Eqs.~(\ref{eqno_6a_VM})  and (\ref{eqno_6b_VM})), one can solve the
constitutive    rate    equation,   written   in   probability   density   space
(Eq.~(\ref{eqno_3}))  to obtain the complete probabilistic description, in terms
of   probability   density   function,  of  evolutionary  stress  response  with
strain (pseudo-time).
% 


One may  note  that  the  probabilistic framework
presented  above  is  a pure constitutive level (point-location
scale)  framework  and  assumes the input soil properties to be
random   variables.   The   spatial   average   (local-average)
probabilistic   constitutive   response,  if  sought  for,  for
example,  in dealing with uncertain spatial variability of soil
properties,   usually   modeled   as   random  fields,  would
necessitate   discretization   of  random  fields  into  random
variables    using    appropriate    technique,   for   example,
Karhunen--Lo\`{e}ve      expansion  \cite{Karhunen:1947,Loeve:1948,book:Ghanem}.   Those   random
variables  could  then be propagated through the point-location
scale   constitutive  framework  presented  above  and  spatial
average  constitutive  response  could  be  assembled  using  a
stochastic  finite  element  technique.  Using polynomial chaos
expansion \cite{Wiener:1938,book:Ghanem}  and
Galerkin  technique,  \citeN{Sett:2009b}  proposed one
such  stochastic finite  element  framework  and  applied the framework in
seismic wave propagation through spatially uncertain stochastic
soil.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{\uppercase{Simulation Results and Discussion}}
\label{RESULTS_AND_DISCUSSION}

In   this   section,  the  constitutive  behavior  of  normally
consolidated,     high    plasticity    clay    is    simulated
probabilistically  using  the  FPKE  approach  described in the
previous section. Elastic--perfectly plastic von Mises material  model is
used for clay. The model requires shear
modulus ($G_{max}$) and undrained shear strength ($s_u$) as
input  soil  parameters.  Both  the  soil parameters are easily
obtainable  through  transformation  from commonly used in-situ tests.
%
In this paper, the above properties are obtained from SPT $N$-value.
%
In this context, it is important to mention that the authors understand
the limitations of using SPT $N$-value for (deterministic) estimation of $s_u$ and
$G_{max}$ for clay. 
%
However, the authors' intent here is to demonstrate the power of a simple
constitutive model, but with uncertain soil
parameters, to simulate the actual response of soil.
%


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\subsection{Quantification of Uncertainties in Input Soil Parameters}
\label{UNCERTAINTY_QUANTIFICATION}

Transformation  from  measured  in-situ properties  to  mechanical  properties  usually 
introduces uncertainty,  which is currently (in  traditional  deterministic analysis)
accounted for by applying engineering judgment. 
%
Alternatively, under the
framework   of   probability  theory,  one  could  quantify  the  transformation
uncertainty by modeling it as a random variable. For example, for alluvial clays
in   Japan,   \citeN{Kulhawy:1999B}  proposed  the  following
relationship between SPT $N$-value and undrained shear strength ($s_u$):
%  
\begin{equation}
\label{eqno_RD1}
s_u  = 0.29 \; p_a \; N^{0.72}
\end{equation}
%
where  $p_a = 101,325~\rm{Pa}$  is  the  atmospheric  pressure.  The above relationship
(Eq.~(\ref{eqno_RD1})),  along  with  the data \cite{Hara:1974} from
which      the      relationship      is     developed,     is     plotted     in
FIG.~\ref{fig:data_ShearStrength}.
%
\begin{figure}[!htbp]
\begin{center}
\includegraphics[width=0.7\textwidth]{Fig01.eps}
\caption{Transformation  relationship  between SPT $N$-value and undrained shear
strength, $s_u$}
\label{fig:data_ShearStrength}
\end{center}
\end{figure}
%
The data-scatter in FIG.~\ref{fig:data_ShearStrength}
represents   knowledge  uncertainty  in   the above transformation
equation  (Eq.~(\ref{eqno_RD1})),  and  under probability theory,
can   be   modeled   as   a   random  variable.  To  this  end,
Eq.~(\ref{eqno_RD1}) can be written as:
%
\begin{equation}
\label{eqno_RD1a}
s_u  = 0.29 \; p_a \; N^{0.72} + \chi
\end{equation}
%
where,  $\chi$  is  a  zero-mean random variable and
represent    the    data-residual   with respect to   the   deterministic
transformation  equation.  The  histogram  of  the  residual is
plotted  in  FIG.~\ref{fig:Histogram_ShearStrength}.  Regarding
the  model  for  the  best-fit  probability density function (PDF), a
Gaussian  distribution  can be ruled out as the histogram is skewed.  
After  trying  few  distributions,  a Pearson IV type distribution, with Pearson
parameters of $0$, $2400$, $-2.75 \times 10^5$, and $9 \times 10^8$, was found to best fit the
residual. 
%
The fitted PDF is shown in FIG.~\ref{fig:Histogram_ShearStrength}.
%
\begin{figure}[!htbp]
\begin{center}
\includegraphics[width=0.7\textwidth]{Fig02.eps}
\caption{Histogram  of  the  residual  (w.r.t  the  deterministic transformation
equation) undrained strength, along with fitted probability density function}
\label{fig:Histogram_ShearStrength}
\end{center}
\end{figure}
%
%

Similarly,  for  transformation  between SPT $N$-value and Young's modulus ($E$)
for    alluvial    clays    in  Japan, \citeN{Kulhawy:1999B} 
proposed   a transformation equation, which can be written in probabilistic form
as
%
\begin{equation}
\label{eqno_RD2}
E = 19.3 \; p_a \; N^{0.63} + \chi
\end{equation}
%
FIG.~\ref{fig:data_YoungsModulus}    shows   
experimental  data \cite{Ohya:1982} along with the deterministic transformation
equation  that in this case represents the  mean trend. 
%
\begin{figure}[!htbp]
\begin{center}
\includegraphics[width=0.7\textwidth]{Fig03.eps} \\
\caption{Transformation relationship between SPT $N$-value and pressure-meter Young's modulus, $E$ }
\label{fig:data_YoungsModulus}
\end{center}
\end{figure}
%
The scatter with respect to the mean trend
(deterministic  transformation equation), plotted as histogram,
is  shown in FIG.~\ref{fig:Histogram_YoungsModulus}. 
%
\begin{figure}[!htbp]
\begin{center}
\includegraphics[width=0.7\textwidth]{Fig04.eps}
\caption{Histogram of the residual (w.r.t the deterministic transformation equation) Young's modulus, along with fitted probability density function}
\label{fig:Histogram_YoungsModulus}
\end{center}
\end{figure}
%
A zero-mean  Gaussian random variable with a standard deviation
of     4041.8     kPa     is     found     to     best     fit
(FIG.~\ref{fig:Histogram_YoungsModulus})  the scatter with respect to the
deterministic   equation.  
%
The  above  standard  deviation  is
obtained using maximum likelihood technique. 

In this context, one may note that
the  penetration  test  is  a high strain test (to the order of
10\%;  refer to Figure 65 in FHWA Geotechnical Engineering Circular No. 5 \cite{FHWA:2002})  and, the corresponding estimated
modulus  (Eq.~\ref{eqno_RD2})  is a high-strain modulus. 
%
Hence,  to estimate the corresponding low-strain modulus (needed as the input to
the  von  Mises  material  model, used in this paper), it is multiplied by $17.25$, assuming $(17.25-1)/17.25=94$~\%
reduction  in modulus at 10\% strain, following \citeN{Idriss:1990} and presuming
that  the  reduction  pattern  of Young's modulus follows the same that of shear
modulus.
%
For  example,  at
$N=15$,  a  mean  high-strain  Young's modulus of $10.735$~MPa is
predicted   by  Eq.~(\ref{eqno_RD2}).  
%
The  corresponding  mean
low-strain    Young's   modulus   used   in   this   paper   is
$10.735$~MPa $\times 17.25 = 185$~MPa. 
%
%  
The  multiplying  factor  (low  strain correction factor) for standard deviation
should,  physically,  be  less  than  that  of the mean. This is because, as the
soil   is  sheared  (or,    in    other    words,    as  soil  plastifies),  the
micro-structure of soil changes and  as  a  result,  our  knowledge uncertainty on it
increases.   This   was   experimentally  observed  by  \citeN{Stokoe:2004}. 
Probabilistic  simulations, published elsewhere by the authors \cite{Sett:2010a}, also show such increase
in uncertainty with strain.
%
Hence,  in  absence of experimental data for clay, following the probabilistic
$G/G_{max}$  versus  shear  strain  curve,  suggested  by  \citeN{Stokoe:2004}, a
multiplying  factor  (low  strain  correction factor) of $[1-(0.2375-0.05)/0.05]
\times 17.25 = 10.7$ is used for standard deviation.
%
%
\noindent At  $N$=15,  a  standard  deviation  of  $4.04$~MPa  was
estimated, using  maximum  likelihood  technique, for high-strain
Young's   modulus.  
%
The  corresponding  standard  deviation  of
low-strain       Young's       modulus,       then,       becomes
$4.04 \times 10.7 = 43.2$~MPa.
%
By assuming  undrained condition, one could assume Poisson's ratio to  be  equal
to  0.5 (deterministic) and could transform the Young's modulus to shear modulus
as 
%
\begin{equation}
\label{eqno_RD3}
G_{max}=\frac{E_{max}}{2(1+\nu)}
\end{equation} 
%
One may note that, as the above equation (Eq.~(\ref{eqno_RD3})) that relates elastic
shear  modulus  and  elastic Young's modulus is linear, the elastic  shear  modulus  
($G_{max}$)  would  also be a Gaussian distribution.
%
Hence, the
statistical  properties  of  the elastic shear modulus ($G_{max}$) can easily be
obtained  using  standard  techniques.  For example, at $N=15$, the mean and the
standard   deviation   of   $G_{max}$   are   $185{\rm MPa}/(2(1+0.5))  =  61.6$~MPa  and
$43.2{\rm MPa}/(2(1+0.5)) = 14.4$~MPa respectively.
%
In this context, it is important to emphasize that the above estimation
of the low-strain correction factors would become unnecessary if
small-strain  shear  modulus  ($G_{max}$) is measured  directly   
from   geophysical   tests   or  estimated through
direct correlation   of   geophysical  test-measured  properties (for
example   shear   wave   velocity,   with  SPT  $N$-value).  The
transformation  equations  between SPT $N$-value and shear wave
velocity,  reported  in  the  literature \cite{Hasancebi:2007,Jafari:2002,Pitilakis:1999,Imai:1977} have not been used in this  
paper due to lack of reported data points for a meaningful statistical 
analysis.



In  addition to the transformation uncertainty, discussed above,
soil    properties  also  include  significant   testing
uncertainties.  
%
For  example,  in  SPT, the testing uncertainty
arises  from  equipment,  procedure  and  operator errors. \citeN{Kulhawy:1999A}
proposed typical range of COV
for SPT as 15-45\%. In this paper, an equivalent of 45\% COV is
added to the undrained shear strength ($s_u$) and 15\% COV is added
to the elastic shear modulus ($G_{max}$) to account for SPT testing
uncertainties.  Larger uncertainty is used for 
undrained  shear  strength  ($s_u$) because the shear strength is not unique but depends on
many factors, e.g. direction  of loading, strain rate, boundary conditions, stress
level,  and sample disturbance effects \cite{Ladd:1991,Mayne:2007}.


Uncertain  spatial  variability  represents  the    other    important source of
uncertainty in soil property. 
%
This uncertainty is present because soil properties are  measured
at few locations and then 
extrapolated/interpolated to all (some) other points
of the soil continuum.
%
In  other  words, in  'estimating'  soil properties
between  two  adjacent  boreholes,  uncertain spatial
variability is incurred. 
%
This uncertain spatial variability  is traditionally accounted for  by  applying
engineering judgment.
%
Probability   theory,  on  the  other  hand, deals  with  uncertain  spatial
variability  through  random  field  modeling \cite{Baecher:1993,Fenton:1999A,Fenton:1999B}.
%
Random  field  modeling characterizes the uncertain spatial variability in terms
of standard deviation and correlation structure.
%
The  standard  deviation  is  usually added  to  the uncertainties arising from transformation equation  
and  testing method, while the  correlation structure can be
accounted  for, among others, through stochastic elastic-plastic finite element
method  \cite{Sett:2009b,Sett:2010c}.
%
In  this  paper,  the  uncertain  spatial  variability  has  not been explicitly
accounted  for  as  the  focus of this paper is on point-location (constitutive)
behavior.
%
However,  one  may  note  that the  data  of  SPT $N$-value versus
undrained   shear   strength  (FIG.~\ref{fig:Histogram_ShearStrength})  and  SPT
$N$-value versus
Young's modulus (FIG.~\ref{fig:data_YoungsModulus}) contain some effects of spatial variability
as  SPT is performed at approximately every $30$~cm (1 foot) and the blow counts
obtained in such way represent  average values over that length. 
%


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Simulation of  $G/G_{max}$ and Damping Behavior}
\label{GOVERGMAX_DAMPING}

The above described uncertain data set is used to analyze, using
the probabilistic elastic--plastic constitutive framework (described in Section~\ref{PROBABILISTIC_FRAMEWORK}), 
the undrained cyclic (shear) behavior of clay.
%
Of particular interest is the performance evaluation of a  simple, elastic--perfectly plastic von Mises material model, 
but extended into probability space.

\subsubsection{Probabilistic elastic--plastic stress--strain response}

FIG.~\ref{fig:HysteresisLoop_01}(a) shows the mean shear stress
versus    shear    strain    hysteresis    loop  for  an undrained clay, simulated using
probabilistic von Mises elastic-perfectly plastic material model.
%
The input to the model were the statistics of the soil properties -- elastic shear  modulus, $G_{max}$ and undrained shear strength, $s_u$ -- 
corresponding to SPT $N$-value of 15.
%
The uncertain clay material is cyclically sheared to $ \pm 1.026$\% shear strain. 
%
The
cyclic evolution of standard deviation of shear stress is shown
in   FIG.~\ref{fig:HysteresisLoop_01}(b).   
%
\begin{figure}[!htbp]
\begin{center}
\includegraphics[width=0.47\textwidth]{Fig05a.eps}
\hspace*{0.5truecm}
\includegraphics[width=0.43\textwidth]{Fig05b.eps}
~~~~~~~~~~~~~~~~~~~~~(a)~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~(b)
\caption{Simulated  hysteresis  shear  stress  versus shear strain loop at $\pm$
1.026\% shear strain: (a) mean and (b) standard deviation behavior}
\label{fig:HysteresisLoop_01}
\end{center}
\end{figure}
%
Results shown in Figures~\ref{fig:HysteresisLoop_01}(a)~and~(b) are obtained by
solving    the   Fokker--Planck--Kolmogorov   equation (FPKE; Eq.~(\ref{eqno_3}),  with advection and diffusion coefficients
given  by  Eqs.~(\ref{eqno_6a_VM}) and (\ref{eqno_6b_VM}))  numerically,  with  appropriate
initial  and  boundary  conditions. The solution to the FPKE, the evolutionary PDF of shear stress with shear strain, is then integrated by standard
techniques   to  obtain  the  evolutionary  mean  and  standard
deviation behavior.
%
The   details  of  the  solution  technique  for governing
FPKE can be found in \citeN{Sett:2010a}.
%


It is important to  note   that     simulation   results  shown  in
FIG.~\ref{fig:HysteresisLoop_01}      are  obtained  using  elastic -- perfectly
plastic   von     Mises  material model and  require only two probabilistic soil
parameters     (their  probability  distribution)  --    elastic  shear  modulus
($G_{max}$) and undrained shear strength ($s_u$). 
%
If  probability distribution of material parameters were neglected and only mean
values were used (thus simplifying to deterministic von Mises
elastic--perfectly plastic model) simple bi-linear response would result. 
%
Such (deterministic) bi-linear response is  also shown in FIG.~\ref{fig:HysteresisLoop_01}(a).
%

In FIG.~\ref{fig:HysteresisLoop_01}(a), it is also interesting to observe that the 
mean response (of the full probability response described by the stress PDF) 
is  non--linear even at very small strain, although, the deterministic
model    assumes  linearity till yielding and then behaves as perfectly
plastic material. 
%
Such  nonlinear  mean  response  is  due to the uncertainty  in the
yield   stress,   as    there  is  always  a probability  (however  small)  that  
elastic--plastic response starts at a very small strain.
%
In addition to that, there also exist a probability that material is elastic
at strains past  the (mean) yield point, and since the mean solution is an ensemble average of
all    the    possibilities,  such  probable  elastic influence is extended into
plastic range as well.  
%
One  can  visualize  this probabilistic yielding effect by observing that within a
laboratory  specimen  (considered generally as a representative
volume    element  (RVE; \citeNP{local-34})),  each  of  large (infinite)  number  of particle
contacts has different  and  unknown  yield strengths.
%
%
At  a  given  strain, some of those particle contacts might be elastic 
while others might be fully   yielding.
%
What  is   observed  in  any laboratory experiment is the ensemble
average    (mean)  behavior of all the particle contacts. Similar conclusion was
developed by \citeN{Einav:2008}, using probabilistic micro-mechanical simulation.
%
It may be noted
that the point-location scale constitutive simulation presented
in this paper doesn't account for the scale effect. 
%
Such  scale  effects could be accounted for by quantifying the uncertain spatial
variability (for example,  through random field modeling) of soil and accounting
for  it  in  our  simulation. This can be done, for  example, through stochastic
elastic--plastic finite  element method  in obtaining local-average constitutive
behavior.




%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsubsection{Evolution of secant shear modulus}

Elastic--plastic constitutive simulation, described above, is used to obtain the evolution of 
secant shear modulus with shear strain.
%
The deterministic evolution of the secant shear modulus is shown in FIG.~\ref{fig:G}(a).
%
The deterministic shear modulus remains constant,  equal  to
$G_{max}=61.6$~MPa      until         $\approx  0.3$\%    strain,  representing
deterministic yield point, before suddenly dropping after yield point.
%
FIG.~\ref{fig:G}(a) also shows  mean  and  mean$\pm$standard  deviation  of the evolutionary 
(probabilistic) secant shear modulus.
%
\begin{figure}[!htbp]
\begin{center}
\includegraphics[width=0.441\textwidth]{Fig06a.eps}
\hspace*{0.5truecm}
\includegraphics[width=0.459\textwidth]{Fig06b.eps}
~~~~~~~~~~~~~~~~~~~~~(a)~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~(b)
\caption{Simulated  (a) probabilistic reduction and (b) evolution of coefficient
of variation (COV) of secant shear modulus with cyclic shear strain.}
\label{fig:G}
\end{center}
\end{figure}
%
Compared    with  the  deterministic  evolution  of  secant  shear 
modulus, the mean solution predicts a realistic reduction with cyclic shear strain.
%  %
The  region  between  mean and mean$\pm$standard  deviation contains the
most likely values of evolutionary shear modulus.
%
Coefficient   of  Variation  (COV),  which    can  also  be  used  to  visualize
uncertainty, is shown in FIG.~\ref{fig:G}(b).
%
The  initial  COV  of secant shear modulus was [($14.4$~MPa $+$ $0.15$ $\times$
$61.6$ MPa)$/61.6$~MPa]$\times 100$\% = $38.3$\%. It was calculated from the
mean and standard deviation values for $G_{max}$, obtained earlier 
as  $61.6$~MPa  and $14.4$~MPa, respectively. The second term in the numerator ($0.15$ $\times$
$61.6$ MPa) represents the contribution of the testing uncertainty, which was assumed to
have a COV of 15\%.
%
%
It     is    interesting    to    observe that  COV of
secant shear modulus increases with cyclic shear strain.
%
This  increase  in  uncertainty  comes  from  the  fact  that  as  the  material
plastifies,
%(breaking of internal structure between soil particle), 
this simple two parameter
model becomes less and less accurate. In other words, more
detailed investigation of the soil (micro) structure is needed and more advanced
modeling technique needs to be used if one wishes to reduce
such uncertainty.



The  above  probabilistic  evolution  of  secant  shear modulus
(FIG.~\ref{fig:G})  is  shown  in FIG.~\ref{fig:GoverGmax} in a
more  common  form,  in  terms of variation of $G/G_{max}$ versus
shear  strain.  
%
\begin{figure}[!htbp]
\begin{center}
\includegraphics[width=0.7\textwidth]{Fig07.eps}
\caption{Simulated probabilistic $G/G_{max}$ behavior}
\label{fig:GoverGmax}
\end{center}
\end{figure}
%
It is important to note that, in  FIG.~\ref{fig:GoverGmax}, the  
normalizations  of  evolutionary mean and mean$\pm$standard
deviation  are done by dividing each of those by the mean of elastic
shear  modulus (Mean[$G_{max}$]). 
%
In other words, the upper and
lower  limits  of  normalized  secant  shear  modulus, shown in
FIG.~\ref{fig:GoverGmax} represent   (Mean[G]$\pm$Standard
Deviation[G])/Mean[$G_{max}]$, rather than $G/G_{max} \pm$standard
deviation.  
%
The  probabilistic  evolution of
material   damping   ratio   versus  shear  strain  is  shown  in
FIG.~\ref{fig:Damping}.   
%
The mean damping ratio, shown in FIG.~\ref{fig:Damping}
is  obtained from the hysteresis loop of mean shear stress versus
shear   strain.
%
Likewise, the upper and lower bounds of damping
ratio    in FIG.~\ref{fig:Damping}, are obtained from the hysteresis
loops of
mean$\pm$standard  deviation  of shear stress versus shear strain.
%
%
%
\begin{figure}[!htbp]
\begin{center}
\includegraphics[width=0.7\textwidth]{Fig08.eps}
\caption{Simulated probabilistic material damping behavior}
\label{fig:Damping}
\end{center}
\end{figure}

In  both   FIGs.~\ref{fig:GoverGmax}  and
\ref{fig:Damping},   the   deterministic   solutions  are  also
plotted. Though the deterministic solutions fail to predict 
realistic soil behavior, probabilistic solutions,   even  with  the  simplest
elastic--perfectly  plastic  model,   are comparable to the
experimental  observations  reported  in  the  literature.  For
example,   the  probabilistic  $G/G_{max}$  and  damping  ratio
curves,    presented  
in FIGs..~\ref{fig:GoverGmax} and
\ref{fig:Damping},  compared well with the experimental data
reported   by  \citeN{Vucetic:1991}  and  \citeN{Stokoe:2004}  for  high
plasticity clay.


In  addition  to  modulus reduction, the probabilistic approach
also  captures  modulus  degradation  when the clay material is
cyclically  sheared  repeatedly. The simulated
hysteresis loops when the clay material is sheared repeatedly to       $\pm$0.1026\%       strain,      is      shown      in
FIG.~\ref{fig:HysteresisLoop_02}.  
%
\begin{figure}[!htbp]
\begin{center}
\includegraphics[width=0.45\textwidth]{Fig09a.eps}
\hspace*{0.5truecm}
\includegraphics[width=0.45\textwidth]{Fig09b.eps} 
~~~~~~~~~~~~~~~~~~~~~(a)~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~(b)
\caption{Simulated hysteresis shear stress versus shear strain loop, when sheared repeatedly at $\pm$ 0.1026\% strain: (a) mean and (b) standard deviation behavior; First four
cycles are shown}
\label{fig:HysteresisLoop_02}
\end{center}
\end{figure}
%
Only the first  four loops are
shown  in  FIG.~\ref{fig:HysteresisLoop_02}  for  clarity.  The
absolute  values of mean and standard deviation of secant shear
modulus     after     each     cycle     are     plotted     in
FIGs.~\ref{fig:ModulusDegradation}(a) and (b), respectively. 
%
\begin{figure}[!htbp]
\begin{center}
\includegraphics[width=0.45\textwidth]{Fig10a.eps} 
\hspace*{0.75truecm}
\includegraphics[width=0.45\textwidth]{Fig10b.eps}
~~~~~~~~~~~~~~~~~~~~~(a)~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~(b)
\caption{Simulated probabilistic degradation of shear modulus, when sheared repeatedly at $\pm$ 0.1026\% strain: (a) mean and (b) standard deviation behavior}
\label{fig:ModulusDegradation}
\end{center}
\end{figure}
%
The
mean  shear  modulus  degrades  8\% after 10 cycles at 0.1026\%
strain.  The  rate  of degradation of mean secant shear modulus
is  higher  initially,  but stabilizes as the number of cycles
increases.  The  standard deviation of secant shear modulus, on
the  other  hand,  increases  (275\% increase after 10 cycles at
0.1026\%  strain)  with  number  of  cycles.  It, however, also
stabilizes  as the  number  of  cycles increases. 
%
%


The  explanation  for  increased   uncertainty  in  the secant shear modulus  is
based on mechanics. 
%
With  repeated shearing, soil (micro) structure is  continuously disturbed
and hence, our knowledge uncertainty on it increases.
%
In other words, after repeated shearing, simplistic  two-parameter  
elastic-perfectly plastic model, used here, cannot model such changes accurately.
% is not accurate enough to model that.
%
The  elastic-plastic probabilistic solution (advection-diffusion equation) aptly
captures that fact.
%
The diffusion component, which controls the spread of the response (stress)
probability density function, keeps evolving continuously with strain, irrespective
of the direction of loading (shearing) until plasticity is fully mobilized with
100\% probability, when the diffusion coefficient becomes zero.
%
The advection component, on the other hand, controls the translation
of the response (stress) PDF in the stress-strain
domain.
%
This component is a function of loading (shearing) direction, advection coefficient and
initial condition at the beginning of each loading direction, which in turn, 
is a function of the uncertainty present in the system at that state of 
strain/shearing.
%
The  advection  component  hence  gives  rise to the degraded modulus after each
cycle, until plasticity is fully mobilized with 100\% probability.
%
The   modulus  degradation  is, therefore, appearing  as a direct consequence of
probabilistic yielding of material (clay).



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{\uppercase{Conclusions}}

Presented  in  this  paper  was  a  probabilistic  approach  to
constitutive 
%  (point-location scale) 
simulation of undrained clay behavior.  
It    had    been    shown that probabilistic approach allowed for  
not only quantification of our confidence in numerical
prediction,  but also modeling modulus reduction, modulus
degradation and damping behavior with simple elastic-perfectly plastic (two-parameter)
material model.
%
This is particularly significant since in geotechnical engineering practice, due to various
constraints, advanced laboratory tests are rarely  performed,
%
while in--situ tests are usually preferred, data from
which  can  be used to calibrate probabilistic material models, one of which was
presented here.
%
In    particular,   shown  here  was  (probabilistic) calibration of  a  simple,
elastic--perfectly  plastic  model  but extended into probability space, by
using number of in--situ tests. Such calibrated probabilistic elastic--perfectly
plastic  model  was then used to predict various aspects of undrained shear behavior of clay.
%
It  was  shown that simulation results compared well with published data (within 
mean $\pm$ standard deviation).
%
More  importantly,  as results  contained  full  PDFs of
the responses, they might have many (other) uses in research and practice.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Acknowledgment}

The work presented in this paper was supported by a grant from Civil, Mechanical and
Manufacturing Innovation program, Directorate of Engineering of the National Science
Foundation, under Award $NSF-CMMI-0600766$ (cognizant program director Dr. Richard
Fragaszy).

% 
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



%
%\\bibliography{SFEM.April2010,SoilProperties,refmech}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

                      
\end{document}
\bye
                                                                                                                                                             

